Author’s response to reviews

**Title:** Laparoscopic Bariatric Surgery Training Program Model: Gastric Bypass.

**Authors:**

- Fabio R. Thuler (fabiothuler@gmail.com)
- Wilson R. Freitas Jr (wrfjr@uol.com.br)
- Elias J. Ilias (eliasjilias@gmail.com)
- Paulo Kassab (paulokassab@terra.com.br)
- Carlos A. Malheiros (camalheiros@gmail.com)

**Version:** 6  **Date:** 28 October 2014

**Author’s response to reviews:**

As requesting:

1. We provided further post-review copyediting.
2. Included consent statement.
3. Specified authors contributions.